Author's response to reviews

Title: Total smoking bans in psychiatric inpatient services: A survey of perceived benefits, barriers and support among staff

Authors:

Paula Wye (Paula.Wye@newcastle.edu.au)
Jenny Bowman (Jenny.Bowman@newcastle.edu.au)
John Wiggers (John.Wiggers@hnehealth.nsw.gov.au)
Amanda Baker (amanda.baker@newcastle.edu.au)
Jenny Knight (Jenny.Knight@hnehealth.nsw.gov.au)
Vaughan Carr (v.carr@unsw.edu.au)
Margarett Terry (Margarett.terry@hnehealth.nsw.gov.au)
Richard Clancy (Richard.Clancy@hnehealth.nsw.gov.au)

Version: 4 Date: 18 June 2010

Author's response to reviews:

The title has been amended in the manuscript document.
Affiliation requirements/addressing have been provided.
Shading has been removed from tables, and bold lettering added.
References have been rechecked and amended, particularly the 'in press' refs updated.
A final check of typography has been completed.
The manuscript has definitely benefitted from this stringent review process - again thank you to reviewers and editors.